Author’s response to reviews

Title: The (cost) effectiveness of procedural sedation and analgesia versus general anaesthesia for hysteroscopic myomectomy, a multicentre randomised controlled trial: PROSECCO trial, a study protocol

Authors:

Julia van der Meulen (juliavandermeulen@gmail.com)
Marlies Bongers (m.bongers@mmc.nl)
Sjors Coppus (Sjors.Coppus@mmc.nl)
Judith Bosmans (j.e.bosmans@vu.nl)
José Maessen (Jose.maessen@mumc.nl)
Katrien Oude Rengerink (koudere2@umcutrecht.nl)
Lucilla Overdijk (l.e.overdijk@olvg.nl)
Celine Radder (c.radder@olvg.nl)
Lucet van der Voet (l.vandervoet@dz.nl)
Nicole Smeets (n.smeets@zuyderland.nl)
Huib van der Vliet (Huib.v.vliet@catharinaziekenhuis.nl)
Wouter Hehenkamp (w.hehenkamp@vumc.nl)
Arentje Manger (pmanger@diakhuis.nl)
Wilbert Spaans (Wilbert.spaans@mumc.nl)
Erica Bakkum (E.A.Bakkum@olvg.nl)
Nicole Horré (nhorree@flevoziekenhuis.nl)
Justine Briët (j.briet@zgt.nl)
Jan Willem van der Steeg (j.v.d.steeg@jbz.nl)
Helen Kok (hskok@alrijne.nl)
Author’s response to reviews:

Dear editor,

Thank you very much for your comments.

We have revised our manuscript according to your comments:

- We have forwarded copies of the funding and ethical approval documents in Dutch (original documents) and translated versions in English to BMCSeriesEditorial@biomedcentral.com.

- In our cover letter and in the funding section we have explained that we received funding from The Dutch organisation for Health Research and Development (ZonMW), which is a governmental funding organisation. In the process of receiving this grant, our study protocol was peer-reviewed by ZonMW.

- In our cover letter and in the manuscript we have explained that our study is still ongoing and we have provided the first subject’s enrollment date.

- No articles concerning the results of this study have been published yet.

- We have provided references for most questionnaires listed in the methods section. For the questionnaires we have not provided a reference for, we have uploaded copies of a Dutch and/or English version as additional files.

- We have included a summary of our informed consent procedure at the ethics and consent to participate section.

- We have made the requested editorial changes to the sample size section and the figure file.